Author's response to reviews

Title: Food store owners' and managers' perspectives on the food environment: An exploratory mixed-methods study

Authors:

Clarence C Gravlee (cgravlee@ufl.edu)
P. Qasimah Boston (abarakaq@aol.com)
M. Miaisha Mitchell (mmiaisha@aol.com)
Alan F Schultz (alan.schultz@ufl.edu)
Connie Betterley (cjbetterley@msn.com)

Version: 2
Date: 24 February 2014

Author's response to reviews:

In the revised manuscript, we have added required statements to confirm that we obtained written informed consent from all participants and that we have no competing interests.